Author's response to reviews

Title: Prenatal diagnosis of a rare form of congenital mid-ureteral stricture: a case report and literature revisited

Authors:

Milena Brugnara (brugnara@supereva.it)
Mariangela Cecchetto (mariangela.cecchetto@libero.it)
Riccardo Manfredi (riccardo.manfredi@univr.it)
Zuffante Michele (michelezuffante@yahoo.it)
Vassilios Fanos (vfanos@tiscal_i.it)
Marco Zaffanello (marco.zaffanello@univr.it)
Angelo Pietrobelli (angpie@tin.it)

Version: 3  Date: 10 May 2007

Author's response to reviews: see over
Dear Dr Annabel Phillips

Senior Assistano Editor
BMC-series Journals
e-mail: editorial@biomedcentral.com
Fax: +44 (0) 20 7631 9923


Dear Dr Annabel Phillips

I’m providing to send the paper to an native English language professional. As soon the manuscript is been corrected I’ll send you again for further consideration.

I’ll send you next week and by fax the written consent obtained from the parents.

Thank you for your assistance.

Best regards.

Marco Zaffanello
Si autorizza il Dr Marco Zaffanello e colleghi alla pubblicazione su BMC urology degli accertamenti clinici e radiologici eseguiti su mia figlia Meghi Elena.

Tale pubblicazione deve comunque avere scopi scientifici. Per nessun motivo il nome o altro riferimento anagrafico, che possa far risalire ad essa, deve comparire nel testo.

In fede

I GENITORI

…………………………………….

Dr Marco Zaffanello and colleagues are authorized to the publication on the journal BMC urology of the clinical and radiological assessments of my daughter Meghi Elena.

Such publication must mast have only scientific scopes. For no reason the name or other identifying reference must appear in the text.

In faith

THE PARENTS

……………………………………..